Author's response to reviews

Title: Impact of pharmacist interventions on adherence and measurable patient outcomes among depressed patients: a randomised controlled study

Authors:

Khalaf A Aljumah Dr. (khalafaljumaah@yahoo.com)
Mohamed Azmi Hassali Assoc. Prof. Dr. (alamalpharmacy@yahoo.com)

Version: 6 Date: 22 July 2015

Author's response to reviews: see over
Reviewer 1 report:

This sentence is not clear: After 6 months, patients in the IG showed significant differences than those in the CG in medication adherence, treatment satisfaction, general overuse beliefs, and specific concern beliefs.
It would be clearer to say: After 6 months, patients in the IG had significantly more favorable medication adherence, treatment satisfaction, general overuse beliefs, and specific concern beliefs.

I thank the reviewer for their comments.
I follow the comment from reviewer.
Also I changed reference style from Harvard style to Vancouver to meet the journal style.